Oral vitamin B12 versus intramuscular vitamin B12 for vitamin B12 deficiency (Review)

Vidal-Alaball J, Butler C, Cannings-John R, Goringe A, Hood K, McCaddon A, McDowell I, Papaioannou A



This is a reprint of a Cochrane review, prepared and maintained by The Cochrane Collaboration and published in *The Cochrane Library* 2009, Issue 1

http://www.thecochranelibrary.com



TABLE OF CONTENTS

EADER	1
BSTRACT	1
LAIN LANGUAGE SUMMARY	2
ACKGROUND	2
BJECTIVES	3
ΙΕΤΗΟDS	3
ESULTS	5
ISCUSSION	8
UTHORS' CONCLUSIONS	9
CKNOWLEDGEMENTS	9
EFERENCES	10
HARACTERISTICS OF STUDIES	14
ATA AND ANALYSES	20
PPENDICES	20
/HAT'S NEW	23
ISTORY	23
ONTRIBUTIONS OF AUTHORS	23
ECLARATIONS OF INTEREST	24
OURCES OF SUPPORT	24
NDEX TERMS	24

[Intervention Review]

Oral vitamin B12 versus intramuscular vitamin B12 for vitamin B12 deficiency

Josep Vidal-Alaball¹, Christopher Butler², Rebecca Cannings-John², Andrew Goringe³, Kerry Hood², Andrew McCaddon⁴, Ian McDowell⁵, Alexandra Papaioannou⁶

¹National Public Health Service for Wales, Swansea, UK. ²Department of Primary Care and Public Health, School of Medicine, Cardiff University, Cardiff, UK. ³Department of Haematology, Cardiff and Vale NHS Trust, Cardiff, UK. ⁴Department of General Practice, Wales College of Medicine, Cardiff University, Wrexham, UK. ⁵Department of Medical Biochemistry and Immunology, Cardiff University, Cardiff, UK. ⁶Department of Medicine, McMaster University, Hamilton, Canada

Contact address: Josep Vidal-Alaball, National Public Health Service for Wales, 36 Orchard Street, Swansea, Wales, SA1 5AQ, UK. Josep.Vidal-Alaball@nphs.wales.nhs.uk.

Editorial group: Cochrane Metabolic and Endocrine Disorders Group. **Publication status and date:** Edited (no change to conclusions), published in Issue 1, 2009. **Review content assessed as up-to-date:** 30 December 2004.

Citation: Vidal-Alaball J, Butler C, Cannings-John R, Goringe A, Hood K, McCaddon A, McDowell I, Papaioannou A. Oral vitamin B12 versus intramuscular vitamin B12 for vitamin B12 deficiency. *Cochrane Database of Systematic Reviews* 2005, Issue 3. Art. No.: CD004655. DOI: 10.1002/14651858.CD004655.pub2.

Copyright © 2009 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

ABSTRACT

Background

Vitamin B12 deficiency is common and rises with age. Most people with vitamin B12 deficiency are treated in primary care with intramuscular vitamin B12 which is a considerable source of work for health care professionals. Several case control and case series studies have reported equal efficacy of oral administration of vitamin B12 but it is rarely prescribed in this form, other than in Sweden and Canada. Doctors may not be prescribing oral formulations because they are unaware of this option or have concerns regarding effectiveness.

Objectives

To assess the effectiveness of oral vitamin B12 versus intramuscular vitamin B12 for vitamin B12 deficiency.

Search methods

Searches were undertaken of *The Cochrane Library*, MEDLINE, EMBASE and Lilacs. The bibliographies of all relevant papers identified using this strategy were searched. In addition we contacted authors of relevant identified studies and Vitamin B12 research and pharmaceutical companies to enquire about other published or unpublished studies and ongoing trials.

Selection criteria

Randomised controlled trials (RCTs) examining the use of oral or intramuscular vitamin B12 to treat vitamin B12 deficiency.

Data collection and analysis

All abstracts or titles identified by the electronic searches were independently scrutinised by two reviewers. When a difference between reviewers arose, we obtained and reviewed a hard copy of the papers and made decisions by consensus. We obtained a copy of all preselected papers and two researchers independently extracted the data from these studies using piloted data extraction forms. The whole group checked whether inclusion and exclusion criteria were met, and disagreement was decided by consensus. The methodological quality of the included studies was independently assessed by two researchers and disagreements were brought back to the whole group and resolved by consensus.

Main results

Two RCT's comparing oral with intramuscular administration of vitamin B12 met our inclusion criteria. The trials recruited a total of 108 participants and followed up 93 of these from 90 days to four months. High oral doses of B12 (1000 mcg and 2000 mcg) were as effective as intramuscular administration in achieving haematological and neurological responses.

Authors' conclusions

The evidence derived from these limited studies suggests that 2000 mcg doses of oral vitamin B12 daily and 1000 mcg doses initially daily and thereafter weekly and then monthly may be as effective as intramuscular administration in obtaining short term haematological and neurological responses in vitamin B12 deficient patients.

PLAIN LANGUAGE SUMMARY

Oral vitamin B12 versus intramuscular vitamin B12 for vitamin B12 deficiency

Vitamin B12 deficiency can cause anaemia and neurological complications. Vitamin B12 is rarely prescribed in the oral form in most countries. Two randomised controlled studies were included in this review. The trials recruited a total of 108 participants and followed up 93 of these from 90 days to four months. The evidence derived from these limited studies suggests that high oral doses of B12 (1000 mcg and 2000 mcg) could be as effective as intramuscular administration in achieving haematological and neurological responses.

BACKGROUND

Description of the condition

Vitamins are a group of substances required for effective human metabolism. Under normal circumstances it is essential that they are present in a person's diet because, with few exceptions, the human body is unable to manufacture them. Vitamins often work together with enzymes, co-factors, and other substances. Vitamin B12 (cobalamin) is necessary for the development of red blood cells, growth, and nervous system maintenance. The only dietary sources of vitamin B12 are animal products, such as eggs, fish and meats. Daily requirements are age-related; the recommended dietary amounts for adults and infants are 1,5 mcg and 0.4 mcg respectively (FAO/WHO 1988, DH 1991).

Vitamin B12 is absorbed in the terminal ileum. This absorption is almost entirely dependent upon intrinsic factor (IF), a glycoprotein secreted by parietal cells situated in the mucosa of the stomach. IF binds to Vitamin B12 and the complex is transported across the cell membrane bound to another glycoprotein called transcobalamin. The most common cause of vitamin B12 deficiency is autoimmune pernicious anaemia, a condition that carries an increased risk of gastric cancer. In pernicious anaemia absorption is impaired due to IF deficiency arising from autoimmune destruction of parietal cells. Other common causes of vitamin B12 deficiency include gastrectomy, ileal resection, pancreatic insufficiency and malabsorption syndromes including Crohn's disease and coeliac disease. Other less common causes include use of drugs such as biguanides (metformin), antacids (proton pump inhibitors and H2 receptors antagonists), aminoglycoside antibiotics, colchicines, and rarely malabsorption due to gastrointestinal bacterial overgrowth and infestation. Pure nutritional deficiency is rare and usually occurs only in strict vegans (EVM 2002).

The true prevalence of vitamin B12 deficiency in the general population is still uncertain but there is evidence that it is common, especially amongst the elderly. The incidence of vitamin B12 deficiency increases with age, probably due to the fact that elderly people are more likely to suffer from food-cobalamin malabsorption. This malabsorption is caused primarily by gastric atrophy but also by chronic carriage of Helicobacter pylori, long term ingestion of metformin and proton pump inhibitors and increased chances of having gastric surgery (Andres 2004). The prevalence of vitamin B12 deficiency in the elderly varies substantially in different studies, largely because of inconsistent diagnostic criteria for vitamin B12 deficiency. Prevalence figures between 1.5% and 15% have been reported (Clarke 2004; Figlin 2003; Pennypacker 1992; Rajan 2002; van Walraven 1999). As well as varying with age, prevalence also varies with gender and ethnic group; elderly men are more likely to have low vitamin B12 levels than elderly women and the prevalence of vitamin B12 deficiency is higher in Europe than in the USA (Lindenbaum 1994; Pennypacker 1992). Vitamin B12 deficiency causes symptoms very similar to folate deficiency: megaloblastic anaemia, chronic tiredness, loss of appetite and mood disturbance. If left untreated, serious neurological and neuropsychiatric complications occur. Vitamin B12 deficiency has also been linked with an increased risk of myocardial infarction and stroke; however this link has not been proved to be causal and it is sustained on the relation between high homocysteine levels and cardiovascular events (Nygard 1997, Carmel 2003). The diagnosis of vitamin B12 deficiency is based mainly on blood measurements of serum vitamin B12, complemented with second lines tests including total homocysteine and methylmalonic acid levels which are metabolic indicators of vitamin B12 deficiency. Studies have indicated that some patients, mainly elderly, can present with neuropsychiatric symptoms in the absence of haematological abnormalities (Lindenbaum 1988).

Description of the intervention

Vitamin B12 (cobalamin) was first isolated in its cyano- form in 1948 (Rickes 1948; Smith 1948), and is now widely used for the treatment of vitamin B12 deficiency. Vitamin B12 replacement has been traditionally administered intramuscularly. However several case control and case series studies have since suggested equal efficacy and safety of the oral route (Chalmers 1958; Ross 1954; Spies 1949). The mechanism for this oral route is most probably that free vitamin B12 can be absorbed both passively (without binding to IF) as well as actively (following binding to IF) in the terminal ileum. Passive diffusion accounts for 1.2% of total absorption with a bioavailability unaffected in patients with pernicious anaemia or gastro-duodenal surgical resection (Berlin 1968; Berlin 1978). High doses of oral vitamin B12 (e.g. 1,000 micrograms daily) may be able to produce adequate absorption of vitamin B12 even in the presence of IF deficiency and therefore be an alternative to the intramuscular route in many patients.

Despite availability in most countries and a very safe track record, vitamin B12 is rarely prescribed in the oral form. In Sweden, however, in the year 2000, oral vitamin B12 accounted for 73% of the total vitamin B12 prescribed (Norberg 2001; Nilsson 2005). Oral vitamin B12 is also widely used in Canada. Possible reasons for doctors not prescribing oral formulations include unawareness of this option or concerns regarding unpredictable absorption (Lederle 1991; Lederle 1998). In the UK, oral vitamin B12 is not currently available on NHS prescription in high dose formulations.

Most vitamin B12 deficient individuals in the UK and other countries are treated with intramuscular vitamin B12. Intramuscular vitamin B12 can be administered in two different forms: cyanocobalamin and hydroxocobalamin. In some countries hydroxocobalamin has completely replaced cyanocobalamin as first choice for vitamin B12 therapy because it is retained in the body longer and can be administered at intervals of up to three months (BNF 2004). Intramuscular injections are a "considerable source of work" for health care professionals, mainly general practitioners and community nurses (Middleton 1985), and can cause significant pain in thin patients (Elia 1998). While serious adverse reactions are rare, injections can be dangerous in anticoagulated patients. There is little difference in the cost of oral versus intramuscular therapy when the medication alone is considered. However, intramuscular administration often involves a special trip to a health facility or a home visit by a health professional to administer the injection (Lederle 1991). Oral treatment could therefore save considerable Health Service resources.

Archie Cochrane, working in South Wales, was one of the first to identify that substantial financial savings could be made from more evidence-based prescribing of vitamin B12. His focus was on excessive prescription of vitamin B12 (Cochrane 1971). More recently, a Canadian study (van Walraven 2001) has estimated that converting patients aged over 65 years on B12 replacement from the intramuscular to oral form could save between \$ 2.9 and \$ 17.6 million (2.5 and 15.3 million Euro) over five years in Ontario alone.

Why it is important to do this review

There have been several case control and non-randomised studies assessing the effectiveness of oral vitamin B12 over the last five decades (e.g. Andres 2001; Andres 2003a; Berlin 1968; McIntyre 1960; Spies 1949). We identified a narrative review (Lane 2002) but did not identify any systematic review comparing oral vitamin B12 with intramuscular administration.

OBJECTIVES

To assess the effects of oral vitamin B12 versus intramuscular vitamin B12 for vitamin B12 deficiency.

METHODS

Criteria for considering studies for this review

Types of studies

Randomised controlled trials examining the use of oral versus intramuscular vitamin B12 to treat vitamin B12 deficiency. No minimum trial duration or post intervention follow-up restriction.

Exclusions:

• studies examining the role of vitamin B12 in the prevention of cardio-vascular diseases, because the dose of vitamin B12 used in those studies may be different from the dose used to treat vitamin B12 deficiency and the majority of patients included in these studies are not vitamin B12 deficient;

• studies of patients with primary folate deficiency, because the concomitant use of folate might confound the metabolic outcome measures;

• studies of patients with end-stage renal disease or on haemodialysis, because renal disease might confound the metabolic outcome measures.

Types of participants

Study participants with low serum vitamin B12 levels, meeting criteria for vitamin B12 replacement therapy. We used a cut-off point of 180 pmol/L (or 240 pg/mL) as threshold serum level for vitamin B12 deficiency. Although there has been some controversy regarding the cut-off point to define vitamin B12 deficiency, values between 120 and 180 pmol/L have been consistently used and 180 pmol/L is a value that ensures a wider inclusion of studies.

Types of interventions

Oral vitamin B12 versus intramuscular vitamin B12.

Types of outcome measures

Primary outcomes

• serum vitamin B12 levels;

• clinical signs and symptoms of vitamin B12 deficiency (e.g., depression, tiredness, paralysis, dementia).

Secondary outcomes

- haemoglobin and mean corpuscular volume (MCV);
- total homocysteine and serum methylmalonic acid levels;costs;

• adverse effects of oral and intramuscular vitamin B12 treatment;

- acceptability to patients;
- quality of life (ideally measured with a validated instrument).

Search methods for identification of studies

Electronic searches

We did not have any language restrictions. We searched the following databases:

• *The Cochrane Library* (Issue 1, 2005), including the Cochrane Database of Systematic Reviews, the Cochrane Central

Register of Controlled Trials (CENTRAL) and the Database of Reviews of Effectiveness (DARE);

- MEDLINE (up to December 2004);
- EMBASE (up to December 2004);
- Lilacs www.bireme.br (up to December 2004).

To identify ongoing trials we searched:

• The National Research Register (UK) (http://www.updatesoftware.com/national);

- Current Controlled Trials (http://www.controlled-trials.com);
- National Institutes of Health (USA) (http://

clinicalstudies.info.nih.gov).

Our search strategy was based on the strategy described in the Cochrane Reviewers' Handbook (Optimal Search Strategy for RCTs. Appendix 5c). The search strategy was used for searches of MEDLINE Ovid Web and adapted for other electronic databases. For the detailed search strategy see Appendix 1.

It was agreed that if during our search we identified any additional key words of relevance, we would adapt our strategy accordingly and re-run the search.

Searching other resources

The bibliographies of all relevant papers identified by this strategy were searched for additional studies.

We contacted (by e-mail or letter) the authors of relevant studies, experts in the field and Vitamin B12 manufacturers to enquire about additional published or unpublished studies, ongoing trials and to obtain additional references.

Data collection and analysis

Selection of studies

All abstracts or titles identified by the electronic searches were independently scrutinised by two researchers (JVA examined all the abstracts and divided the abstracts between CCB, RCJ and KH to provide a second review). When uncertainty arose, or when there were differences between these reviewers, we obtained and reviewed the hard copies of papers and made decisions by consensus.

The whole group checked whether inclusion and exclusion criteria were met and disagreement was resolved by consensus.

Data extraction and management

We obtained a copy of all selected papers. JVA extracted data from all the papers and another researcher (either AMC or RCJ) also extracted data from each paper using piloted data extraction forms.

Assessment of risk of bias in included studies

The methodological quality of the included studies were independently assessed by two researchers (CCB and RCJ) and disagreements brought back to the whole group for discussion.

We used the scheme described in the Cochrane Collaboration Handbook for assessing methodological quality. This scheme uses the criteria specified by Schulz and by Jadad (Jadad 1996; Schulz 1995) and involves assessing studies for:

• publication bias

• selection bias - a) was the randomisation procedure adequate? b) was the allocation concealment adequate?

- performance bias a) were the patients and people
- administering the treatment blind to the intervention? • attrition bias - a) were withdrawals and dropouts

completely described? b) was analysis by intention-to-treat?

• detection bias - were outcome assessors adequately blinded to the intervention patients received?

Based on these criteria a 3-point rating scale was used, with the following grading:

A - all quality criteria met: low risk of bias (plausible bias is unlikely to alter the results seriously)

B - one or more of the quality criteria only partly met: moderate risk of bias (plausible bias raises some doubt about the results)

C - one or more criteria not met: high risk of bias (plausible bias seriously weakens confidence in the results)

Data synthesis

We planned to perform meta-analysis if studies were sufficiently similar in terms of inclusion criteria, settings, treatment, follow up assessment and quality. We aimed to evaluate equivalence between the oral and intramuscular routes of vitamin B12 administration. We planned to report serum concentrations of vitamin B12, homocysteine and methylmalonic acid and sources of clinical outcome of both routes of administration, oral and parenteral, giving mean differences and confidence intervals. For other outcome measures, relative risk with 95% confidence intervals or effect sizes will be reported.

We intended to assess sources of heterogeneity in order to determine whether variation observed in the results could be explained by chance alone. We intended to use the chi-squared test and the I-square test to examine evident heterogeneity between the trials (Higgins 2002). If significant heterogeneity was evident (P < 0.1), we could have attempted to determine reasons through subgroup and sensitivity analyses.

If in the future adequate data become available we will use this method of analysis.

Subgroup analysis and investigation of heterogeneity

We planned subgroups for the following:

- Newly diagnosed patients
- Patients already established on intramuscular treatment
- People over 65 years of age
- Patients who have had stomach or small bowel resection
- Vegetarians
- Patients taking anti-ulcer medication over long periods
- Patients with dementia
- Patients with acquired immunodeficiency syndrome

Sensitivity analysis

When relevant, we planned to repeat the analysis:

- excluding studies published in abstract form only;
- taking account of study quality, as specified abov;
- excluding any unusually long term or large studie;.

• excluding studies using the following filters: source of funding (industry versus other), country.

We planned to test the robustness of the results by repeating the analysis using different measures of effect size (risk difference, odds ratio etc.) and different statistical models (fixed and random effects models).

RESULTS

Description of studies

See: Characteristics of included studies; Characteristics of excluded studies.

Data extraction

Two groups of researchers independently extracted data from the two randomised studies that met the inclusion criteria using piloted data extraction forms. The characteristics are described in the table Characteristics of included studies. We did not extract data from non-randomised studies or studies that clearly did not meet the inclusion criteria. The whole group checked all the relevant studies in relation to inclusion and exclusion criteria.

Language

Most of the publications were written in English (88%), but we found seven papers that we felt warranted closer scrutiny written in other languages: Italian (Aguzzi 1985; Migliorini 1979), French (Brassinne 1974; De La Fourniere 1997; Kaltenbach 2003), Danish (Bastrup-Madsen 1973) and Czech (Bezdickova 1997).

JVA and MF independently assessed the Italian studies. JVA and DP assessed the three French studies. BN and AMH independently assessed the Danish study. JVA and CCB assessed the English

abstracts of the Czech study and this study was excluded on the basis of the abstract alone.

Results of the search

Our initial search included studies published before 2003. After the protocol was published, we decided to include the studies published during 2003 and 2004. After manually removing all duplicates, we pre-selected 797 abstracts identified by searches. We selected 41 abstracts for which the full paper was obtained and two studies met the inclusion criteria.

Handsearching

Searching the bibliographies of the selected articles and other relevant studies identified another 15 studies for which we obtained the full text.

A total of 56 studies were thus selected for review of the full paper.

Additional search strategies

Clinical experts (Dr Norberg, Dr Burr, Dr Dempsey and Dr Lederle) authors of relevant identified studies (Dr Andres, Dr Nyholm, Dr Bolaman and Dr Allen) and Vitamin B12 manufacturers (Goldshield Pharmaceuticals, Link Pharmaceuticals and Celltech Group) were contacted to enquire about other published or unpublished studies, ongoing trials and to obtain additional references. No additional studies were identified in this way.

Interrater agreement

Each stage of selection of studies was followed by discussions with the group. The process resulted in agreement in the pre-selection stage, the quality of the assessment, and in the data extraction phase. We did not use kappa statistic (Cohen 1960) because during the pre-selection process we obtained hard copies of any study independently identified by researchers. When assessing the methodological quality of the included studies kappa statistic was not calculated as just two studies met the inclusion criteria.

Included studies

Two randomised control trials (Bolaman 2003; Kuzminski 1998) fulfilled our inclusion criteria. Characteristics of these studies are shown in Characteristics of included studies

Study types

Two studies compared oral administration of vitamin B12 versus intramuscular administration of vitamin B12. The two studies were randomised controlled trials.

Participants

The trials recruited a total of 108 participants and followed up 93 of these from 90 days to four months. 52 participants were female (56%) and 41 were male (44%). Participants in both studies had low serum vitamin B12 levels and both were set in outpatient hospital clinics. The first study was carried out in USA by Kuzminski and colleagues (Kuzminski 1998) and the second in Turkey by Bolaman and colleagues (Bolaman 2003).

Kuzminski and colleagues recruited 38 patients with a mean age of 72 years for the oral group and 71 years for the intramuscular group. 28 of these patients had conditions that may be associated with malabsorption from the gut (including seven with pernicious anaemia and three with ileal resection), although patients with inflammatory bowel disease and coeliac disease appear not to have been included.

Bolaman and colleagues recruited 70 patients with a mean age of 60 years for the oral group and 64 years for the intramuscular group. 35 of these patients had conditions affecting the ileum that may be associated with malabsorption from the gut. Again, however, patients with inflammatory bowel disease and coeliac disease appear not to have been included.

Patient inclusion and exclusion criteria in individual studies are outlined in Characteristics of included studies.

Interventions

The two studies compared oral administration of vitamin B12 versus intramuscular administration of vitamin B12. The duration of the intervention was the same as the duration of the follow up. The dose of oral vitamin B12 used by Kuzminski and colleagues was 2,000 mcg (Kuzminski 1998); Bolaman and colleagues used 1,000 mcg (Bolaman 2003). The dose of intramuscular vitamin B12 was 1,000 mcg in both studies and it was administered by nurses.

Outcome measures

Several primary outcomes measures were reported in the studies: • Serum vitamin B12

- Haemoglobin
- Mean Corpuscular Volume (MCV)
- White Blood Cells (WBC) •
- Platelet count
- Methylmalonic Acid
- Homocysteine
- Folate
- •
- Serum creatinine • Fasting serum gastrin
- Parietal and intrinsic factor antibodies
- Mini-Mental State Examination
- Neurologic assessment

Only one of the studies (Bolaman 2003) in addition reported about tolerability of medication (assessed by a haematologist and patient interviews), and costs, assessed by the authors using cost of the study drug and the injections.

Excluded studies

Fifty-four studies were excluded. Reasons for exclusion of studies are given in Excluded studies. The most common reason for exclusion was a non-randomised trial design (57%) or the study did not meet intervention criteria (35%).

Risk of bias in included studies

Further details of the methodological quality of the included studies can be found in Appendix 2.

Allocation

One of the studies (Kuzminski 1998) did not describe attempts to conceal the assignment of participants.

Patients were randomised using the block randomization method described by Altman (Altman 1999) to allocate participants (Bolaman 2003). The other study (Kuzminski 1998) stated that they used a statistical package for randomisation but did not provide further details.

Blinding

The two studies (Bolaman 2003; Kuzminski 1998) were not blinded, as patients knew whether they were taking oral or intramuscular medication since no placebo was used.

Incomplete outcome data

In the study by Kuzminski and colleagues (Kuzminski 1998), 38 participants were randomised but five participants were excluded from the final analysis because they were judged to have primary folate deficiency rather than Vitamin B12 deficiency. The final analysis included 18 participants in the oral group and 15 randomised to receive intramuscular treatment. Three participants randomised to receive intramuscular vitamin B12 had a total of seven samples excluded from the analysis because of an error in taking the specimen. However the samples were judged valid for assay of serum metabolites. The authors did not analyse according to intention to treat.

In the study by Bolaman and colleagues (Bolaman 2003), 70 participants were enrolled in the study but 10 were excluded because failure to appear to follow up. The final analysis included 26 participants in the oral group and 34 randomised to receive intramuscular treatment. The authors did not analyse according to intention to treat.

Other potential sources of bias

Sample size calculation

None of the studies reported a sample size calculation.

Length of follow-up

The length of follow up ranged from 90 days to four months. We consider this as insufficient because of the long biological half-life of body stores of vitamin B12. This is estimated to be more than 30 month (Basu 1996).

Effects of interventions

Two studies compared oral and intramuscular vitamin B12: Kuzminski and colleagues (Kuzminski 1998) reported neurologic and haematologic responses. Four of the eighteen participants randomised to receive oral vitamin B12 and four of the fifteen randomised to receive intramuscular vitamin B12 had a neurological response with a marked improvement or clearing of paresthesias, ataxia, or memory loss. Serum vitamin B12 levels were significantly higher in the oral (643 +/- 328 pg/mL) compared to the intramuscular group (306 +/- 118 pg/mL) at 2 months (P<0.001). The difference was even greater at four months (1,005 +/- 595 vs. 325 +/- 165 pg/mL; P< 0.0005). Serum methylmalonic acid concentrations decreased to < 3 SD above the normal range in all participants except one in the oral and two in the intramuscular group. Mean concentrations of the metabolites were not significantly different between the oral and the intramuscular groups, except at four months, when the value was higher in the intramuscular group (P< 0.05). Elevated serum total homocysteine decreased to 3 SD above the normal range in most participants, but the decrease was over four months in the oral group and during the first month in the intramuscular group. However, in two patients in each group the response was not optimal.

Bolaman and colleagues (Bolaman 2003) reported neurologic and haematologic responses. Both groups receiving oral or intramuscular vitamin B12 reported improvements of cognitive function, sensory neuropathy and vibration sense but the difference between both groups was not statistically significant. Serum vitamin B12 levels increased in those receiving oral and those receiving intramuscular vitamin B12 for 90 days. The authors reported a statistically significant difference between day 0 and day 90 within both groups (P< 0.001) but did not analyse differences between both groups.

Heterogeneity and subgroup analyses

It was not possible to investigate heterogeneity or to do subgroup analysis as the four studies differed markedly in terms of interventions.Sensitivity analysesSensitivity analyses were not possible

due to the small number of studies that met the inclusion criteria. We could have included more studies if we had considered participants with normal vitamin B12 levels at study entry but the objective of the review was to look at the use of vitamin B12 for the treatment of vitamin B12 deficiency. Adverse effects, acceptability and quality of lifeOnly one of the studies mentioned adverse effects and tolerability (Bolaman 2003). In this study, the authors stated that the oral treatment was better tolerated but they did not report on how they assessed tolerability. None of the studies gave an account of the effect of the intervention in the quality of life of participants. CostsCosts were reported in one of the included trials (Bolaman 2003). In the trial, the cost of the treatment was \$ 80 (66 Euro) per participant in the oral group and \$ 220 (184 Euro) per participant in the intramuscular group (P< 0.001). Authors did not report which economic analysis was performed.

DISCUSSION

Summary of main results

We identified two randomised controlled trials directly relevant to the study question. The follow up period ranged between 90 days to four months. A total of 108 patients with vitamin B12 deficiency participated in these studies. After consideration of study populations, treatments, outcome measures and follow up, we concluded that meta-analysis was not appropriate. A major consideration was heterogeneity of oral treatment regimes and diverse eligibility for study entry.

The authors of the two studies comparing oral with intramuscular administration of vitamin B12 concluded that oral vitamin B12 was as effective as the intramuscular treatment. Kuzminski and colleagues (Kuzminski 1998) concluded that 2000 mcg daily of oral cyanocobalamin was as effective or even superior to 1000 mcg of the same medication administered intramuscularly every month. Bolaman and colleagues (Bolaman 2003) concluded that 1000 mcg of oral or intramuscular vitamin B12 daily for 10 days followed by once weekly for four weeks and once monthly for life was equally effective to treat patients with megaloblastic anaemia. Neither study was blinded. However, the conclusions were based on haematological as well as neurological responses to both treatments and laboratory assessments were likely to have been blinded. A crucial aspect our study addresses is whether or not patients with conditions that might cause malabsorption may be safely treated with oral vitamin B12 in primary care. This was difficult to ascertain because in the trials numbers are small, follow up short and trials did not include patients with common conditions that might interfere with absorption in the terminal ileum such as Crohn's disease, coeliac disease or ulcerative colitis. Three patients with ileal resection and seven with pernicious anaemia were included in one of the studies. Despite these cautions, we did find evidence of a satisfactory short term response to oral vitamin B12 replacement even in patients with some conditions, mainly affecting the upper gastrointestinal tract that may be associated with malabsorption.

Overall completeness and applicability of evidence

The studies included older participants with a mean age ranging from 60 to 72. This is consistent with the higher prevalence of Vitamin B12 deficiency amongst the elderly (Rajan 2002; van Walraven 1999). However, the regimen described in the two studies, will be particularly challenging for the elderly with dementia. None of the studies was conducted in Primary Care where most of the patients with vitamin B12 deficiencies are treated; this makes the generalisability of the results more difficult. Another factor affecting generalisability is that the four studies used different treatment regimes and used strict and numerous exclusion criteria.

The study conducted by Kuzminski and colleagues (Kuzminski 1998) used injectable cyanocobalamin and not hydroxocobalamin. The bioavailability of the water solution of cyanocobalamin used is much lower than that of more usual injectable preparations such as hydroxocobalamin. For example, Glass and co-workers showed that 69% of a dose of 1000 mcg of cyanocobalamin given as injection was recovered in the urine over 72 hours. The corresponding percentage after an injection of 1000 mcg of hydroxocobalamin was 27% (Glass 1961). The amount of retained B12 would therefore be quite low in the injection group in this particular study. This probably explains why oral B12 appeared to be more effective than this particular injectable preparation.

In one of the studies using higher doses of oral vitamin B12 (Bolaman 2003), the authors reported that no adverse events occurred. This is consistent with previous studies (Hathcock 1991). The outcomes were limited and the length of follow up was probably too short to demonstrate effectiveness of the oral treatment, this makes the applicability of the results difficult.

Potential biases in the review process

This body of evidence has serious limitations as it includes only two open studies with relatively short follow up periods (90 days and four months) and small number of participants (n = 38 and n = 70) with some attrition. In addition, there are methodological limitations to both studies identified by our search strategy. In one of the studies (Kuzminski 1998) the method of randomisation it is not described. Allocation concealment was adequate in just one study (Bolaman 2003). Intention to treat analysis and blinding was not performed or mentioned in any of the studies. We did not carry out meta-analysis because of heterogeneity of designs, dosages and participants.

Agreements and disagreements with other studies or reviews

We identified many studies assessing the effectiveness of oral vitamin B12, but they were not randomised. The studies, dating from the early 50's to recently, were mainly before and after studies and all show that patients with vitamin B12 deficiency respond to oral vitamin B12 replacement therapy in clinical and/or laboratory terms (Andres 2001; Andres 2003a; Berlin 1968; Brody 1959; Chalmers 1958; Conley 1952; Conley 1955; Hemsted 1958; Kondo 1998 ; McIntyre 1960; Reisner 1955; Spies 1949; Unglaud 1955; Ungley 1950; Verhaeverbeke 1997; Waife 1963). However, it is not clear in most of these studies how many included patients suffered from conditions that may cause malabsorption.

AUTHORS' CONCLUSIONS

Implications for practice

The limited evidence identified in this systematic review shows that *high doses* of oral vitamin B12 (2000 mcg) daily are as effective as the intramuscular administration (Kuzminski 1998) in obtaining haematological and neurological responses in patients with vitamin B12 deficiency. High doses of oral vitamin B12 (1000 mcg) initially daily and thereafter weekly and then monthly are also as effective as intramuscular vitamin B12 (Bolaman 2003). The included studies also showed limited evidence for a satisfactory haematological, biochemical and clinical short term response for oral B12 replacement in some patients with conditions associated with malabsorption.

Current clinical practice in UK and in most countries is to prescribe vitamin B12 in the intramuscular form for the treatment of vitamin B12 deficiency. This has been the norm for the last 50 years despite several non-randomised studies in the early 1950's demonstrating satisfactory responses to oral treatment and the fact that there is considerable experience in Sweden in using oral vitamin B12. In 1998, the study by Kuzminski et al. (Kuzminski 1998) was the first randomised controlled trial to show that in achieving a satisfactory neurological, haematological and biochemical response, daily high doses of oral vitamin B12 were as effective or even more effective than intramuscular vitamin B12 when treating patients with vitamin B12 deficiency.

Generalised oral vitamin B12 treatment might benefit many patients in terms of fewer visits to health carers and reduced discomfort associated with injections. Nursing time would be released for treating other patients. However, adherence and monitoring will remain important considerations, regardless of route of administration.

Implications for research

The above evidence has not been sufficient to change practice in most countries. A further large, pragmatic trial in a primary care setting is needed to determine whether oral vitamin B12 is effective in patients with major common cases of malabsorption and to provide additional evidence regarding cost effectiveness.

In addition, clinicians are often concerned about patient preferences regarding treatment route. General Practitioners commonly report that patients receiving intramuscular vitamin B12 are reluctant to reduce the frequency of their injections or change from intramuscular to oral vitamin B12 replacement, despite having serum vitamin B12 levels several times the normal range. Many patients receive vitamin B12 injections more often than the recommended three month interval (Cochrane 1971; Fraser 1995; van Walraven 1999). Some individuals started on vitamin B12 for 'tiredness', but with normal haematological and biochemical parameters, report feeling better on the injections and are reluctant to stop them. This suggests that intramuscular B12 may carry additional psychotropic effects for patients, exceeding those associated with normalisation of serum vitamin B12 serum levels. In a recent qualitative and quantitative study, 73% of patients were willing to try oral vitamin B12 and of those who tried the oral therapy 71% wished to permanently switch. They mentioned travel inconveniences as reasons for preferring the oral route (Kwong 2005). However, in another non-randomised control trial, researchers attempted to convert patients in primary care from intramuscular to oral vitamin B12 replacement but only half agreed to participate in the study (Nyholm 2003). Possible psychotropic effects are important because they may make General Practitioners and patients reluctant to change to oral therapy, despite evidence that this route is as effective and probably cheaper than the intramuscular route. A deep understanding of patient's preferences, explaining possible psychotropic effects of intramuscular vitamin B12 therapy, and developing effective ways of reaching shared decisions with patients is probably the way forward.

Further research is therefore needed to avoid perpetuating oral vitamin B12 replacement as one of "medicine's best kept secrets" (Lederle 1991).

ACKNOWLEDGEMENTS

• Mirella Longo (School of Care Science. University of Glamorgan, Pontypridd) assessed the Italian articles.

• Diana Pasterfield (Department of General Practice. Wales College of Medicine, Wrexham) assessed the French studies.

• Dr Bo Norberg (Department of Internal Medicine, University Hospital of Northern Sweden, Sweden) and Dr Anne-

Mette Hvas (Department of Clinical Biochemistry, University of Aarhus, Denmark) assessed the Danish study.

• Dr Bolaman, Dr Burr, Dr Nyholm, Dr Dempsey, Dr Lederle, Dr Andres and Dr Allen were contacted to enquire about other published or unpublished studies, ongoing trials and to obtain additional references.

• Members of the Cardiff University Vitamin B12 Group, including Prof Mark Worwood, Dr Michael Burr and Mr Dick Ellis supported the process.

REFERENCES

References to studies included in this review

Bolaman 2003 {published data only}

Bolaman Z, Kadikoylu G, Yukselen V, Yavasoglu I, Barutca S, Senturk T. Oral Versus Intramuscular Cobalamin Treatment in Megaloblastic Anemia: A Single-Center, Prospective, Randomized, Open-Label Study. *Clinical Therapeutics* 2003;**25**(12):3124–34.

Kuzminski 1998 {published data only}

Kuzminski AM, Del Giacco EJ, Allen RH, Stabler SP, Lindenbaum J. Effective treatment of cobalamin deficiency with oral cobalamin. *Blood* 1998;**92**(4):1191–8.

References to studies excluded from this review

Adachi 2000 {published data only}

Adachi S, Kawamoto T, Otsuka M, Todoroki T, Fukao K. Enteral vitamin B12 supplements reverse postgastrectomy B12 deficiency. *Annals of Surgery* 2000;**232**(2):199–201.

Aguzzi 1985 {published data only}

Aguzzi F, Dispari V, Contardi M, et al. Treatment of psychic regression of the elderly: A double-blind clinical trial [Su un tratamento della patologia psico–involutiva dell'anziano. Ricerca clinica in doppio cieco]. *Giornale di Gerontologia* 1985;**33**(7):593–603.

Altay 1999 {published data only}

Altay C, Cetin M. Vitamin B12 absorption test and oral treatment in 14 children with selective vitamin B12 malabsorption. *Pediatric Hematology and Oncology* 1999;**16** (2):159–63.

Andres 2001 {published data only}

Andres E, Kurtz JE, Perrin AE, Maloisel F, Demangeat C, Goichot B, et al.Oral cobalamin therapy for the treatment of patients with food-cobalamin malabsorption. *American Journal of Medicine* 2001;**111**(2):126–9.

Andres 2003 {published data only}

Andres 2003a {published data only}

Andres E, Kaltenbach G, Noel E, Noblet-Dick M, Perrin A-E, Vogel T, et al.Efficacy of short-term oral cobalamin therapy for the treatment of cobalamin deficiencies related to food-cobalamin malabsorption: A study of 30 patients. *Clinical and Laboratory Haematology* 2003;25(3):161–6.

Andres 2003b {published data only}

Andres E, Kaltenbach G, Noel E, Noblet-Dick M, Perrin A_E. Short-term oral cobalamin therapy for food-related cobalamin malabsorption. *Annals of Pharmacotherapy* 2003; **37**(2):301–2.

Andres(2) 2003 {published data only}

Anonymous 1997 {published data only}

Anonymous. Many more treatment options available. *Cleveland Clinic Journal of Medicine* 1997;**64**(9):497–502.

Bastrup-Madsen 1973 {published data only}

Bastrup-Madsen P. Comparison between the durations of the effects of the vitamin B12 preparations Betolvex and Vibeden [Summenligning mellen varigheden af virkingen af B12–vitaminpraeparaterne Belolvex of Vibeden]. *Ugeskrift for Laeger* 1973;**135**(38):2037–41.

Berlin 1968 {published data only}

Berlin H, Berlin R, Brante G. Oral treatment of pernicious anemia with high doses of vitamin B12 without intrinsic factor. *Acta Medica Scandinavica* 1968;**184**(4):247–58.

Bezdickova 1997 {published data only}

Bezdickova D, Vacha K, Schneiderka P. Biochemical monitoring of vitamin B12 substitution therapy [Biochemicke monitorovani substitucni terapie vitaminem B12]. *Klinicka Biochemie a Metabolismus* 1997;**5**(3):180–4.

Boddy 1968 {published data only}

Boddy K, King P, Mervyn L, Macleod A, Adams JF. Retention of cyanocobalamin, hydroxocobalamin, and coenzyme B12 after parenteral administration. *The Lancet* 1968;**2**(7570):710–2.

Brassinne 1974 {published data only}

Brassinne A. Evaluation of the simultaneous administration of radioactive vitamin B12 bound to the intrinsic gastric factor and of free vitamin B12 in the diagnosis of pernicious anemia [Evaluation de l'administration simultanee de vitamine B12 radio–active liee au facteur intrinseque gastrique et de vitamine B12 radio–active libre pour le diagnostic de l'anemie pernicieuse]. *Journal Belge de Radiologie* 1974;**57**(1):1–7.

Brody 1959 {published data only}

Brody E, Stren S, Wasserman L. Treatment of Pernicious Anemia by oral administration of Vitamin B12 without added Intrinsic Factor. *The New England Journal of Medicine* 1959;**260**(8):361–7.

Brody 1967 {published data only}

Brody EA. Failure of long-term oral vitamin B12 therapy to replenish body B12 stores. *American Journal of Clinical Nutrition* 1967;**20**(5):443–5.

Chalmers 1958 {published data only}

Chalmers JN, Shinton NK. Absorption of orally administered vitamin B12 in pernicious anaemia. *The Lancet* 1958;**2**:1298–1302.

Chalmers 1965 {published data only}

Chalmers JN, Shinton NK. Comparison of hydroxocobalamin and cyanocobalamin in the treatment of pernicious anaemia. *The Lancet* 1965;**2**(7426):1305–8.

Conley 1952 {published data only}

Conley C, Green T, Hartmann R, Krevans J. Prolongued Treatment of Pernicious Anemia with Vitamin B12. *The American journal of medicine* 1952:284–93.

Conley 1955 {published data only}

Conley C, Krevans J. New developments in the diagnosis and treatment of pernicious anemia. *Annals of Internal Medicine* 1955;**43**:758–66.

De La Fourniere 1997 {published data only}

De La Fourniere F, Ferry M, Cnockaert X, Chahwakilian A, Hugonot-Diener L, Baumann F, et al.Vitamin B12 deficiency and dementia: A multicenter epidemiologic and therapeutic study. Preliminary therapeutic trial [Deficience en vitamine B12 et etat dementiel: etude epidemiologique multicentrique et therapeutique. Essai preliminaire]. *Semaine des Hopitaux* 1997;**73**(5-6):133–40.

Ellis 1973 {published data only}

Ellis FR, Nasser S. A pilot study of vitamin B12 in the treatment of tiredness. *British Journal of Nutrition* 1973;**30** (2):277–83.

Glass 1966 {published data only}

Glass GB, Skeggs HR, Lee DH. Hydroxocobalamin. V. Prolonged maintenance of high vitamin B12 blood levels following a short course of hydroxocobalamin injections. *Blood* 1966;**27**(2):234–41.

Hedstrand 1969 {published data only}

Hedstrand H. Urinary excretion of hydroxocobalamine and cyanocobalamine after oral administration of large doses. *Acta Medica Scandinavica* 1969;**186**(6):535–7.

Hemsted 1958 {published data only}

Hemsted E, Mills J. Vitamin B12 in Pernicious Anaemia. Intramuscular or Oral. *The Lancet* 1958;**2**(7060):1302–3.

Hughes 1970 {published data only}

Hughes D, Elwood PC, Shinton NK, Wrighton RJ. Clinical trial of the effect of vitamin B12 in elderly subjects with low serum B12 levels. *British Medical Journal* 1970;1(707): 458–60.

Hvas 2003 {published data only}

Hvas A-M, Juul S, Nexo E, Ellegaard J. Vitamin B-12 treatment has limited effect on health-related quality of life among individuals with elevated plasma methylmalonic acid: A randomized placebo-controlled study. *Journal of Internal Medicine* 2003;**253**(2):146–52.

Kaltenbach 2003 {published data only}

Kaltenbach G, Noblet-Dick M, Andres E, Barnier-Figue G, Noel E, Vogel T, et al.Early response to oral cobalamin therapy in older patients with vitamin B12 deficiency [Reponse precoce au traitement oral par vitamine B12 chez des sujets ages hypovitaminiques]. *Annales de Medecine Interne* 2003;**154**(2):91–5.

Killander 1968 {published data only}

Killander A, Werner I. Maintenance therapy of pernicious anaemia. Comparison of cyanocobalamin, hydroxocobalamin and a cyanocobalamin-zinc-tannate complex. *Acta Haematologica* 1968;**40**(6):305–14.

Kondo 1998 {published data only}

Kondo H. Haematological effects of oral cobalamin preparations on patients with megaloblastic anaemia. *Acta Haematologica* 1998;**99**(4):200–5.

Kwok 1998 {published data only}

Kwok T, Tang C, Woo J, Lai WK, Law LK, Pang CP. Randomized trial of the effect of supplementation on the cognitive function of older people with subnormal cobalamin levels. *International Journal of Geriatric Psychiatry* 1998;**13**(9):611–6.

Mader 1988 {published data only}

Mader R, Deutsch H, Siebert GK, Gerbershagen HU, Gruhn E, Behl M, et al.Vitamin status of inpatients with chronic cephalgia and dysfunction pain syndrome and effects of a vitamin supplementation. *International Journal for Vitamin & Nutrition Research* 1988;**58**(4):436–41.

Mann 1987 {published data only}

Mann BA, Garry PJ, Hunt WC, Owen GM, Goodwin JS. Daily multivitamin supplementation and vitamin blood levels in the elderly: a randomized, double-blind, placebocontrolled trial. *Journal of the American Geriatrics Society* 1987;**35**(4):302–6.

Martin 1992 {published data only}

Martin DC, Francis J, Protetch J, Huff FJ. Time dependency of cognitive recovery with cobalamin replacement: report of a pilot study. *Journal of the American Geriatrics Society* 1992;**40**(2):168–72.

Mauro 2000 {published data only}

Mauro GL MU, Cataldo P, Brancato G, Letizia G. Vitamin B12 in low back pain: a randomised, double-blind, placebo-controlled study. *European Review for Medical & Pharmacological Sciences* 2000;4(3):53–8.

McIntyre 1960 {published data only}

McIntyre PA HR, Masters JM, Krevans JR. Treatment of pernicious anemia with orally administered cyanocobalamin (vitamin B12). *Archives of Internal Medicine* 1960;**106**: 280–92.

Migliorini 1979 {published data only}

Migliorini A, Licopoli R, Palieri PL. Comparative evaluation of the therapeutic effectiveness of Cobamamide at high doses by intramuscular injection and by iontophoresis. A controlled investigation [Confronto dell'efficacia terapeutica tra la somministrazione per via intramuscolare e via jonoforetica della Cobamamide a forti dosi. Studio controllato]. *Progresso Medico* 1979;**35**(9):399–412.

Naurath 1995 {published data only}

Naurath HJ, Joosten E, Riezler R, Stabler SP, Allen RH, Lindenbaum J. Effects of vitamin B12, folate, and vitamin B6 supplements in elderly people with normal serum vitamin concentrations. *The Lancet* 1995;**346**(8967):85–9.

Nyholm 2003 {published data only}

Nyholm E, Turpin P, Swain D, Cunningham B, Daly S, Nightingale P, et al.Oral vitamin B12 can change our practice. *Postgraduate Medical Journal* 2003;**79**(930):218–9.

Pathy 1972 {published data only}

Pathy MS, Pippen CA, Kirkman S. Free and intrinsic factor bound radioactive cyanocobalamin. Simultaneous administration to assess the significance of low serum vitamin B 12 levels. *Age and Ageing* 1972;1(2):111–9.

Reisner 1955 {published data only}

Reisner E, Weiner L, Schittone M, Henck E. Oral Treatment of Pernicious Anemia with Vitamin B12 without Intrinsic Factor. *The New England Journal of Medicine* 1955;**12**: 502–6.

Rhode 1995 {published data only}

Rhode BM, Tamim H, Gilfix BM, Sampalis JS, Nohr C, MacLean LD. Treatment of vitamin B12 deficiency after gastric surgery for severe obesity. *Obesity Surgery* 1995;**5**(2): 154–8.

Ross 1954 {published data only}

Ross G, Mollin D, Cox E, Ungley C. Hematologic responses and concentration of vitamin B12 in serum and urine following oral administration of vitamin B12 without intrinsic factor. *Blood* 1954;**9**:473–88.

Saltzman 1993 {published data only}

Saltzman JR. Vitamin and trace-element supplementation in the elderly. *Annals of Internal Medicine* 1993;**118**(6 SUPPL).

Seal 2002 {published data only}

Seal EC, Metz J, Flicker L, Melny J. A randomized, double-blind, placebo-controlled study of oral vitamin B12 supplementation in older patients with subnormal or borderline serum vitamin B12 concentrations. *Journal of the American Geriatrics Society* 2002;**50**(1):146–51.

Skouby 1970 {published data only}

Skouby AP. Dosage of hydroxocobalamin for vitamin B12 deficiency [Dosering af hydroksokobalamin (Vibeden)]. *Acta Medica Scandinavica* 1970;**188**(1):31–6.

Slot 1997 {published data only}

Slot WB, Merkus FW, Van Deventer SJ, Tytgat GN. Normalization of plasma vitamin B12 concentration by intranasal hydroxocobalamin in vitamin B12-deficient patients. *Gastroenterology* 1997;**113**(2):430–3.

Spies 1949 {published data only}

Spies T, Stone R, Lopez G, Milanes F, Toca R, Aramburu T. Vitamin B12 by mouth in pernicious and nutritional macrocytic anaemia and sprue. *The Lancet* 1949;**2**:454–6.

Tin 1978 {published data only}

Tin May T, Ma Win M, Khin Sann A, Mya-Tu M. The effect of vitamin B12 on physical performance capacity. *British Journal of Nutrition* 1978;**40**(2):269–73.

Tudhope 1967 {published data only}

Tudhope GR, Swan HT, Spray GH. Patient variation in pernicious anaemia, as shown in a clinical trial of cyanocobalamin, hydroxocobalamin and cyanocobalamin--zinc tannate. *British Journal of Haematology* 1967;**13**(2): 216–28.

Ubbink 1994 {published data only}

Ubbink JB, Vermaak WJH, Van der Merwe A, Becker PJ, Delport R, Potgieter HC. Vitamin requirements for the treatment of hyperhomocysteinemia in humans. *Journal of Nutrition* 1994;**124**(10):1927–33.

Unglaud 1955 {published data only}

Unglaud W, Goldsmith G. Oral Vitamin B12 in the Treatment of Macrocytic Anemias. *Southern Medical Journal* 1955;**48**:261–9.

Ungley 1950 {published data only}

Ungley CC. Absorption of vitamin B12 in pernicious anaemia. I. Oral administration without source of intrinsic factor. *British Medical Journal* 1950;**2**:905–8.

van Asselt 2001 {published data only}

van Asselt DZ, Pasman JW, van Lier HJ, Vingerhoets DM, Poels PJ, Kuin Y, et al. Cobalamin supplementation improves cognitive and cerebral function in older, cobalamin-deficient persons. *Journals of Gerontology Series A-Biological Sciences and Medical Sciences* 2001;**56**(12):M775–9.

Verhaeverbeke 1997 {published data only}

Verhaeverbeke I, Mets T, Mulkens K, Vandewoude M. Normalization of low vitamin B12 serum levels in older people by oral treatment. *Journal of the American Geriatrics Society* 1997;**45**(1):124–5.

Waife 1963 {published data only}

Waife SO JC, Crabtree RE, Grinnan EL, Fouts PJ. Oral vitamin B12 without intrinsic factor in the treatment of pernicious anemia. *Annals of Internal Medicine* 1963;**58**: 810–17.

Additional references

Altman 1999

Altman DG, Bland JM. How to randomise. *BMJ* 1999; **319**:703–4.

Andres 2004

Andres E, Loukili NH, Noel E, Kaltenbach G, Ben Abdelgheni M, Perrin AE, et al.Vitamin B12 (cobalamin) deficiency in elderly patients. *Canadian Medical Association Journal* 2004;**171**(3):251–9.

Basu 1996

Basu TK, Dickerson JWT. *Vitamins in human health and disease*. Wallingford: CAB International, 1996:345.

Berlin 1968

Berlin H, Berlin R, Brante G. Oral treatment of pernicious anemia with high doses of vitamin B12 without intrinsic factor. *Acta Medica Scandinavica* 1968;**184**(4):247–58.

Berlin 1978

Berlin R, Berlin H, Brante G, Pilbrant A. Vitamin B12 body stores during oral and parenteral treatment of pernicious anaemia. *Acta Medica Scandinavica* 1978;**204**(1-2):81–4.

BNF 2004

British Medical Association, Royal Pharmaceutical Society of Great Britain. British National Formulary. British Medical Association, 2004, issue 48:456–7.

Carmel 2003

Carmel R, Green R, Rosenblatt DS, Watkins D. Update on Cobalamin, Folate, and Homocysteine. *Hematology* 2003: 62–81.

Clarke 2004

Clarke R, Grimley Evans J, Schneede J, Nexo E, Bates C, Fletcher A, et al.Vitamin B12 and folate deficiency in later life. *Age and Ageing* 2004;**33**:34–41.

Cochrane 1971

Cochrane AL, Moore F. Expected and observed values for the prescription of Vitamin B12 in England and Wales. *British Journal of Preventive and Social Medicine* 1971;**25**: 147–51.

Cohen 1960

Cohen J. A coefficient of agreement for nominal scales. *Educational and Psychological Measurement* 1960;**20**:37–66.

DH 1991

DH. Dietary Reference Values for Food Energy and Nutrients for the United Kingdom. Report of the Panel on Dietary Reference Values of the Committee on Medical Aspects of Food Policy. *Report on Health and Social Subjects 41*. London: HMSO, 1991.

Elia 1998

Elia M. Oral or parenteral therapy for B12 deficiency. *The Lancet* 1998;**352**(9142):1721–2.

EVM 2002

Expert Group on Vitamin and Minerals. Revised Review on Vitamin B12. Food and Standards Agency UK. URL: http://food.gov.uk/multimedia/pdfs/EVM/00/20/P (accessed 2002).

FAO/WHO 1988

Joint FAO/WHO Expert Consultation. Requirements of vitamin A, iron, folate and vitamin B12. Report of a Joint FAO/WHO Expert Consultation. FAO Food and Nutrition Series. No.23 1988.

Figlin 2003

Figlin E, Chetrit A, Shahar A, Shpilberg O, Zivelin A, Rosenberg N, et al. High prevalences of vitamin B12 and folic acid deficiency in elderly subjects in Israel. *British Journal of Haematology* 2003;**123**:696–701.

Fraser 1995

Fraser RC, Farooqi A, Sorrie R. Use of vitamin B-12 in Leicestershire practices: a single topic audit led by a medical audit advisory group. *BMJ* 1995;**311**:28–30.

Glass 1961

Glass GB, Skeggs, HR, Lee, DH, Jones, EL, Hardy, WW. Hydroxocobalamin. I. Blood levels and urinary excretion of vitamin B12 in man after a single parenteral dose of aqueous hydroxocobalamin, aqueous cyanocobalamin and cyanocobalamin zinc-tannate complex. *Blood* 1961;**18**: 511–21.

Hathcock 1991

Hathcock JN, Troendle GJ. Oral cobalamin for treatment of pernicious anemia?. *Journal of the American Medical Association* 1991;**265**(1):96–7.

Higgins 2002

Higgins JP, Thompson SG. Quantifying heterogeneity in a meta-analysis. *Statistics in Medicine* 2002;**21**(11):1539–58.

Jadad 1996

Jadad AR, Moore RA, Carroll D, Jenkinson C, Reynolds DJ, Gavaghan DJ, et al.Assessing the quality of reports of randomized clinical trials: Is blinding necessary?. *Controlled Clinical Trials* 1996;**17**:1–12.

Kwong 2005

Kwong J, Carr D, Dhalla I, Tomkun D, Upshur R. Oral therapy in the primary care setting: a qualitative and quantitative study of patients perspectives. *BMC Family Practice* 2005;**6**(8).

Lane 2002

Lane LA, Rojas-Fernandez C. Treatment of Vitamin B12-Deficeincy Anemia: Oral Versus Parenteral Therapy. *The Annals of Pharmacotherapy* 2002;**36**(7-8):1268–72.

Lederle 1991

Lederle FA. Oral cobalamin for pernicious anemia. Medicine's best kept secret?. *Journal of the American Medical Association* 1991;**265**(1):94–5.

Lederle 1998

Lederle FA. Oral Cobalamin for Pernicious Anemia: Back from the Verge of Extinction. *Journal of the American Geriatrics Society* 1998;**46**:1125–27.

Lindenbaum 1988

Lindenbaum J, Healton EB, Savage DG, Brust JC, Garret TJ, Podell ER, et al.Neuropsychiatric disorders caused by cobalamin deficiency in the absence of anemia or macrocytosis. *New England Journal of Medicine* 1988;**318**: 1720–8.

Lindenbaum 1994

Lindenbaum J, Rosenberg IH, Wilson PW, Stabler SP, Allen RH. Prevalence of cobalamin deficiency in the Framingham elderly population. *American Journal of Clinical Nutrition* 1994;**60**(1):2–11.

Middleton 1985

Middleton J, Wells W. Vitamin B12 injections: considerable source of work for the district nurse. *British Medical Journal Clinical Research Ed* 1985;**290**(6477):1254–5.

Nilsson 2005

Nilsson M, Norberg B, Hultdin J, Sandstrom H, Westman G, Lokk J. Medical intelligence in Sweden. Vitamin B12: oral compared with parenteral?. *Postgraduate Medical Journal* 2005;**81**:191–3.

Norberg 2001

Norberg B. Oral high-dose cyanocobalamin - a contagious cocept. Rondel. URL: http://www.rondellen.net (accessed 2001); Vol. 8.

Nygard 1997

Nygard O, Nordrehaug JE, Refsum H, Ueland PM, Farstad M, Vollset SE. Plasma homocysteine levels and mortality in patients with coronary artery disease. *New England Journal of Medicine* 1997;**337**:230–6.

Pennypacker 1992

Pennypacker LC, Allen RH, Kelly JP, Matthews LM, Grigsby J, Kaye K. High prevalence of cobalamin deficiency in elderly outpatients. *Journal of the American Geriatrics Society* 1992;**40**:1197–204.

Rajan 2002

Rajan S, Wallace JI, Beresford SA, Brodkin KI, Allen RA, Stabler SP. Screening for cobalamin deficiency in geriatric outpatients: prevalence and influence of synthetic cobalamin intake. *Journal of the American Geriatrics Society* 2002;**50**(4):624–30.

Rickes 1948

Rickes EL, Brink NG, Koniuszy FR, Wood TR, Folkers K. Crystalline vitamin B12. *Science* 1948;**107**:396.

Ross 1954

Ross GIM, Mollin DL, Cox EV, Ungley CC. Hematologic responses and concentration of vitamin B12 in serum and

urine following oral administration of vitamin B12 without intrinsic factor. *Blood* 1954;**9**:473–88.

Schulz 1995

Schulz KF, Chalmers I, Hayes RJ, Altman DG. Empirical evidence of bias: dimensions of methodological quality associated with estimates of treatment effects in controlled trials. *Journal of the American Medical Association* 1995;**273**: 408–12.

Smith 1948

Smith EL. Purification of anti-pernicious anaemia factor from liver. *Nature* 1948;**161**:638.

Spies 1949

Spies TD, Stone RE, Lopez GG, Milanes F, Toca RL, Aramburu T. Vitamin B12 by mouth in pernicious and nutritional macrocytic anaemia and sprue. *Lancet* 1949;**2**: 454–6.

van Walraven 1999

van Walraven C, Naylor CD. Use of vitamin B12 injections among elderly patients by primary care practitioners in Ontario. *Canadian Medical Association Journal* 1999;**161**: 146–9.

van Walraven 2001

van Walraven C, Austin P, Naylor CD. Vitamin B12 injections versus oral supplements. How much money could be saved by switching from injections to pills?. *Canadian Family Physician* 2001;**47**:79–86.

* Indicates the major publication for the study

CHARACTERISTICS OF STUDIES

Characteristics of included studies [ordered by study ID]

Bolaman 2003

Methods	Allocation: randomised Observation period: 90 days				
Participants	Patients with megaloblastic anaemia due to cobalamin deficiency Numbers: 70 patients enrolled, 10 excluded. Outcomes measured for 60 patients. 26 oral group. 34 intramuscular group. 14 patients with atrophic gastritis, 7 with chronic antral gastritis, 7 with chronic pangastritis, 6 with alkaline reflux gastritis and 1 with erosive gastritis Age: Mean age 60 (+/-15) for the oral group and 64 (+/-10) for the intramuscular group Inclusion criteria: - Serum cobalamin level <160pg/mL				
Interventions	1000 mcg cobalamin oral or intramuscular once daily for 10 days. After 10 days, once a week for 4 weeks and after that patients were asked to continue treatment once a month indefinitely				
Outcomes	Primary Outcomes: - Serum cobalamin - Mini-Mental State Examination - Neurologic assessment Additional Outcomes: - Haemoglobin - MCV - WBC - Platelet count - Tolerability - Costs - Side effects not reported				
Notes	Not blinded				

Bolaman 2003 (Continued)

Item	Description				
Allocation concealment?	Unclear]	B - Unclear		
Kuzminski 1998					
Methods	Allocation: randomised Observation period: 4 months				
Participants	Newly diagnosed cobalamin deficient patients Numbers: 38 randomised. Outcomes measured for 33 patients 17 patients with chronic atrophic gastritis, 7 with pernicious anaemia, 3 with ileal resection, 1 with gastric stapling Age: Mean age 72 (+/- 11) for the oral group and 71 (+/- 15) for the intramuscular group Inclusion criteria: - Serum cobalamin level <160pg/mL - Elevation of serum methylmalonic acid, total homocysteine or both metabolites >3 SD's above the mean in normal controls Exclusion criteria: - Location outside immediate geographical area - Incapacity to give informed consent - Refusal to participate - Associated life-threatening illness - Primary Folate deficiency				
Interventions	1000 mcg intramuscular cyanocobalamin on days 1, 3, 7, 10, 14, 21, 30, 60 and 90 2000 mcg oral cyanocobalamin daily for 120 days				
Outcomes	Primary Outcomes: - Serum cobalamin - Neurologic responses Additional Outcomes: - Methylmalonic Acid - Homocysteine - Side effects not reported				
Notes	Not blinded				
Risk of bias					
Item	Authors' judgement]	Description		
Allocation concealment?	Unclear		B - Unclear		

MCV. Mean Corpuscular Volume WBC. White Blood Cells

Characteristics of excluded studies [ordered by study ID]

Study	Reason for exclusion
Adachi 2000	Non-randomised study
Aguzzi 1985	Randomised, placebo-controlled double-blinded study assessing the effectiveness of a product mixture con- taining cyanocobalamin in the treatment of age-dependent psychic regression. No evidence that patients were vitamin B12 deficient
Altay 1999	Non-randomised study
Andres 2001	Non-randomised study
Andres 2003	
Andres 2003a	Non-randomised study
Andres 2003b	Same patients as in Andres 2003, letter in a different publication
Andres(2) 2003	
Anonymous 1997	Review of treatments of diabetes. Non-randomised study
Bastrup-Madsen 1973	Randomised study but comparing intramuscular hydroxocobalamin and intramuscular cyanocobalamin
Berlin 1968	Non-randomised study
Bezdickova 1997	Non-randomised study
Boddy 1968	Cross-over study comparing subjects without vitamin B12 deficiency
Brassinne 1974	Non-randomised study
Brody 1959	Non-randomised study
Brody 1967	Non-randomised study
Chalmers 1958	Non-randomised study
Chalmers 1965	Study comparing intramuscular injections of hydroxocobalamin and cyanocobalamin
Conley 1952	Non-randomised study

Conley 1955	Non-randomised study
De La Fourniere 1997	Randomised, placebo-controlled double-blinded study assessing use of intramuscular vitamin B12 in de- mentia
Ellis 1973	Randomised, cross-over double-blinded study comparing intramuscular hydroxocobalamin and intramus- cular placebo in tiredness
Glass 1966	Non-randomised study
Hedstrand 1969	Non-randomised study
Hemsted 1958	Study comparing intramuscular injections monthly with oral treatment daily. The study does not state whether patients were randomised
Hughes 1970	Randomised, placebo-controlled double blinded study comparing intramuscular hydroxocobalamin with placebo
Hvas 2003	Randomised, placebo-controlled double-blinded study comparing intramuscular vitamin B12 with placebo
Kaltenbach 2003	Non-randomised study
Killander 1968	Non-randomised study
Kondo 1998	Non-randomised study
Kwok 1998	Randomised controlled study comparing intramuscular cyanocobalamin with no treatment
Mader 1988	Randomised, placebo-controlled double-blinded study using an oral multivitamin preparation containing folic acid in headaches. No data available regarding vitamin B12 levels
Mann 1987	Randomised, placebo-controlled blinded study using a multivitamin/mineral preparation containing folic acid. Most of the patients were had not vitamin B12 deficiency
Martin 1992	Non-randomised study
Mauro 2000	Randomised, placebo-controlled double-blinded study studying efficacy of intramuscular vitamin B12 in the treatment of low back pain
McIntyre 1960	Non-randomised study
Migliorini 1979	Randomised controlled double-blinded study looking at the effect of intramuscular vitamin B12 in pain
Naurath 1995	Double-blinded placebo controlled study using an intramuscular multivitamin preparation
Nyholm 2003	Non-randomised study

Pathy 1972	Non-randomised study				
Reisner 1955	Non-randomised study				
Rhode 1995	Study comparing different doses of oral vitamin B12 treatment. No placebo group. No randomisation apparent				
Ross 1954	Non-randomised study				
Saltzman 1993	Unable to locate study as part of a supplement				
Seal 2002	Randomised controlled study comparing different doses of oral cyanocobalamin with oral placebo				
Skouby 1970	Non-randomised study				
Slot 1997	Non-randomised study				
Spies 1949	Non-randomised study				
Tin 1978	Randomised, placebo-controlled double-blinded study comparing the effect of intramuscular cyanocobal- amin and placebo on physical performance				
Tudhope 1967	Randomised study comparing intramuscular hydroxocobalamin and intramuscular cyanocobalamin				
Ubbink 1994	Randomised, placebo-controlled blinded study looking at the effect of vitamin supplementation on hyper- homocysteinemia. Some patients receiving vitamin B12 were not vitamin B12 deficient				
Unglaud 1955	Non-randomised study				
Ungley 1950	Non-randomised study				
van Asselt 2001	Non-randomised study				
Verhaeverbeke 1997	Non-randomised study				
Waife 1963	Non-randomised study				

DATA AND ANALYSES

This review has no analyses.

APPENDICES

Appendix I. Search strategy

Electronic searches

Unless otherwise stated, search terms were free text terms; exp = exploded MeSH: Medical subject heading (Medline medical index term); the dollar sign (\$) stands for any character(s); the question mark (?) = to substitute for one or no characters; tw = text word; pt = publication type; sh = MeSH: Medical subject heading (Medline medical index term); adj = adjacency.

- 1. vitamin\$ b12.tw.
- 2. vitamin\$ b 12.tw.
- 3. cobalamin\$.tw.
- 4. hydroxycobalamin\$.tw.
- 5. cyanocobalamin\$.tw.
- 6. betolvex.tw.
- 7. exp Vitamin B 12/ae, cl, ct, tu [Adverse Effects, Classification, Contraindications, Therapeutic
- Use]
- 8. exp Hydroxocobalamin/ae, tu [Adverse Effects, Therapeutic Use]
- 9. 1 or 2 or 3 or 4 or 5 or 6 or 7 or 8
- 10. exp Vitamin B 12 Deficiency/
- 11. vitamin B12 deficienc\$.tw.
- 12. vitamin B 12 deficienc\$.tw.
- 13. B 12 deficienc\$.tw.
- 14. b12 deficienc\$.tw.
- 15. 10 or 11 or 12 or 13 or 14
- 16. randomized-controlled trial.pt.
- 17. controlled-clinical trial.pt.
- 18. randomized-controlled-trials.sh.
- 19. random allocation.sh.
- 20. double-blind method.sh.
- 21. single-blind method.sh.
- 22. or/16-21
- 23. animal.sh.
- 24. human.sh.
- 25. 23 not 24
- 26. 22 not 25
- 27. clinical trial.pt.
- 28. exp clinical trials/
- 29. (clinic\$ adj25 trial\$).tw.
- 30. ((singl\$ or doubl\$ or trebl\$ or tripl\$) adj (mask\$ or blind\$)).tw.
- 31. Placebos.sh.

32. placebo\$.tw. 33. random\$.tw. 34. research design.sh. 35. (latin adj square).tw. 36. or /27-35 37. 36 not 25 38. 37 not 26 39. exp evaluation studies/ 40. follow-up studies.sh. 41. prospective studies.sh. 42. (control\$ or prospectiv\$ or volunteer\$).tw. 43. cross-over studies.sh. 44. comparative study.sh. 45. or/39-44 46. 45 not 25 47. 46 not (26 or 38) 48. 26 or 38 or 47 49. 9 and 15 and 48 50. from 49 keep 1-627

Appendix 2. Methodological quality of included studies

Study	Randomisa- tion	Alloc. Conceal.	Blinding	N. Randomised	Withdrawals	Intent_to_treat	Follow-up
Kuzminski 1998	Unclear	Unclear	No	38	5	No	4 months
Bolaman 2003	Block ran- domisation method	Probably ade- quate	No	70	10	No	90 days

Appendix 3. Data extraction sheet

Items	
Screener's name	
Study number and name	
Author	
Publication	

Language of Publication
Year of publication
Country
Type of study (case control, case series, article, RCT)
Treatments
Inclusion - Exclusion criteria
Numbers recruited
Numbers randomised
Numbers loss to follow up
Why? (reasons)
Numbers excluded
Why? (reasons)
Withdraws
Why? (reasons)
Outcomes measures (bioquemical, functional)
Randomisation. If yes:
* Length of follow up
* Follow up points
* Side effects
* N. Evaluated in each group
* N. Responded in each group
Quality (Cochrane criteria)
a) low risk of bias (plausible bias is unlikely to alter the results seriously)
b) moderate risk of bias (plausible bias raises some doubt about the results)

c) high risk of bias (plausible bias seriously weakens confidence in the results) Blinding. If yes: * Method of blinding Intention-to-treat analysis Additional treatment (co-interventions) Power calculation (stats)

WHAT'S NEW

Last assessed as up-to-date: 30 December 2004.

Date	Event	Description
5 November 2008	Amended	Converted to new review format.

HISTORY

Protocol first published: Issue 1, 2004

Review first published: Issue 3, 2005

CONTRIBUTIONS OF AUTHORS

JOSEP VIDAL-ALABALL: Coordination, searching, protocol writing, data extraction, interpretation of data, report writing.

CHRISTOPHER C BUTLER: Formulation of study question, searching, protocol writing, data extraction, piloting form for data extraction, study quality assessment, interpretation of data, report writing.

KERENZA HOOD: Statistical advice on study design, study quality assessment, suitability of data for meta-analysis, interpretation of data and report writing.

REBECCA CANNINGS-JOHN: Data extraction, report writing, meta-analysis.

ANDREW MCCADDON: Advice about appropriate biochemical and clinical endpoint, design and piloting of data extraction form, critical appraisal of findings, report writing.

ALEXANDRA PAPAIOANNOU: Clinical appraisal of findings, report writing, expertise on care of older people with long term medication.

IAN MCDOWELL: Biochemical advice. Critical appraisal of findings. Review writing.

ANDREW GORINGE: Haematological advice. Critical appraisal of findings. Review writing.

DECLARATIONS OF INTEREST

We have not received any commercial sponsorship for this review.

Dr Andrew McCaddon is a Scientific Consultant for, and shareholder of, COBALZ LIMITED - a private company developing 'glutathionylcobalamin' as an alternative orally available form of vitamin B12.

SOURCES OF SUPPORT

Internal sources

- Department of General Practice, Wales College of Medicine, Cardiff University, UK.
- National Public Health Service for Wales, Wales, UK.

External sources

• No sources of support supplied

INDEX TERMS

Medical Subject Headings (MeSH)

Administration, Oral; Injections, Intramuscular; Randomized Controlled Trials as Topic; Vitamin B 12 [*administration & dosage]; Vitamin B 12 Deficiency [*drug therapy]; Vitamin B Complex [*administration & dosage]

MeSH check words

Aged; Humans